Dear editor of World J Clin Cases

Thank you for your e-mail

We would like to present our thanks for you for considering our work for publication

We like to thank the reviewers for their effort to improve the quality of the submitted work

We hope the changes that we made will meet your approval

Reviewer 1

1. Rewrite the title so that it clarifies the purpose of writing a letter to the editor.

   Thank you for your comment, a revision was made as requested

2. Concentrate on reformulating the study abstract so that it clarifies the problem and how to solve it.

   Your advice was taken in consideration we thank you for the advice

3. Work on replacing Table No. 1 with linguistic texts that explain its findings.

   The table was replaced with text.

Reviewer #2:

Thank you sir for your comments, wishing you and the rest of the authors all the progress and prosperity.